
%-----------------------------------------------
%           TEISSIER     01/05  LATEX 2e
%-----------------------------------------------
\documentclass{article}
%\usepackage{color}
\usepackage{amsthm,amsmath,amssymb,amscd}
\usepackage[greek,english]{babel}
\def \findem{\smallskip\hfill\vrule width 5 pt height 7 pt depth - 2 pt\smallskip}
%--------------------------------------------------------
\newtheorem{theorem}{Theorem}[section]
\newtheorem{proposition}[theorem]{Proposition}
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{corollary}[theorem]{Corollary}
\theoremstyle{definition}
\newtheorem{definition}[theorem]{Definition}
\newtheorem{exercise}[theorem]{Exercise}
\newtheorem{example}[theorem]{Example}
\theoremstyle{remark}
\newtheorem{remark}[theorem]{Remark}
\newtheorem{Remark}[theorem]{Remarks}
\renewcommand{\qedsymbol}{$\square$}
%\frenchspacing

\def\A{{\cal A}}
\def\Aa{{\mathbf A}}
\def\P{{\cal P}}
\def\Q{{\cal Q}}
\def\Qq{{\mathbf Q}}
\def\R{{\cal R}}
\def\Rr{{\mathbf R}}
\def\B{{\cal B}}
\def\Ss{{\cal S}}
\def\C{{\mathbf C}}
\def\N{{\mathbf N}}
\def\Z{{\mathbf Z}}

\def\limproj{\mathop{\oalign{lim\cr\hidewidth$\longleftarrow$\hidewidth\cr}}}
\def\limind{\mathop{\oalign{lim\cr\hidewidth$\longrightarrow$\hidewidth\cr}}}

\begin{document}
\begin{thebibliography}{10}


\bibitem{H1} Hironaka, Heisuke, On the arithmetic genera and the effective genera of algebraic curves, Mem. Coll. Sci. Univ. Kyoto. Ser. A. Math. {\bf 30} (1957), 177-195.

\bibitem{H2} Hironaka, H., A note on algebraic geometry over ground rings. The invariance of Hilbert characteristic functions under the specialization process. Illinois J. Math. {\bf 2} (1958) 355-366.

\bibitem{H3} Hironaka, H., A generalized theorem of Krull-Seidenberg on parameterized algebras of finite type. Amer. J. Math. {\bf 82} (1960)
831-850.

\bibitem{H4} Hironaka, H., An example of a non-K\"ahlerian complex-analytic deformation of K\"ahlerian complex structures. Ann. of Math. {\sl (2)} {\bf 75} (1962), 190-208.

\bibitem{H5} Hironaka, H., On resolution of singularities (characteristic zero). 1963 Proc. Internat. Congr. Mathematicians (Stockholm, 1962),
507-521 Inst. Mittag-Leffler, Djursholm.

\bibitem{H6} Hironaka, H.; Rossi, H., On the equivalence of imbeddings of exceptional complex spaces. Math. Ann. {\bf 156} (1964), 313-333.

\bibitem{H7} Hironaka, H., Resolution of singularities of an algebraic variety over a field of characteristic zero. I, II. Ann. of Math. {\sl (2)} {\bf 79} (1964), 109-203 and 205-326.

\bibitem{H8} Hironaka, H., A fundamental lemma on point modifications (1965) Proc. Conf. Complex Analysis (Minneapolis, 1964), pp. 194-215 Springer, Berlin.

\bibitem{H9} Hironaka, H., On the equivalence of singularities. I. (1965) Arithmetical Algebraic Geometry (Proc. Conf. Purdue Univ., 1963) pp. 153-200 Harper \& Row, New York.

\bibitem{H10} Iss'sa, Hej, On the meromorphic function field of a Stein variety. Ann. of Math. (2) {\bf 83} (1966) 34-46.

\bibitem{H11} Hironaka, H., On the characters $\nu \sp{*} $ and $\tau \sp{*} $ of singularities, J. Math. Kyoto Univ. {\bf 7} (1967), 19-43.

\bibitem{H12} Hironaka, H., Corrections to: ``On the characters $\nu \sp{*} $ and $\tau \sp{*} $ of singularities", J. Math. Kyoto Univ. 7 1967 325-327.

\bibitem{H13} Hironaka, H., Characteristic polyhedra of singularities. J. Math. Kyoto Univ.
7, (1967) 251--293.

\bibitem{H14} Hironaka, H., Smoothing of algebraic cycles of small dimensions, Amer. J. Math. {\bf 90} (1968) 1-54.

\bibitem{H15} Hironaka, H., On some formal imbeddings. Illinois J. Math. {\bf 12} (1968), 587-602.

\bibitem{H16} Hironaka, H., Matsumura, Hideyuki, Formal functions and formal embeddings. J. Math. Soc. Japan 20 (1968), 52-82.

\bibitem{H17} Hironaka, H., Formal line bundles along exceptional loci, (1969) Algebraic Geometry (Internat. Colloq., Tata Inst. Fund. Res., Bombay, 1968) pp. 201-218 Oxford Univ. Press.

\bibitem{H18} Hironaka, H., Normal cones in analytic Whitney stratifications, Inst. Hautes \'Etudes Sci. Publ. Math. {\bf 36} (1969), 127-138.

\bibitem{H19} Hironaka, H.,  Additive groups associated with points of a projective space. Ann. of
Math. (2), 92 (1970) 327--334. 

\bibitem{H20} Hironaka, H., Certain numerical characters of singularities. J. Math. Kyoto Univ., 10  (1970), 151--187. 

\bibitem{H21} Hironaka, H., Desingularization of complex-analytic varieties, (French) Actes du Congr\`es International des Math\'ematiciens (Nice, 1970), Tome 2, pp. 627-631. Gauthier-Villars, Paris, 1971.

\bibitem{H22} Hironaka, H., Schemes, etc. Algebraic geometry, Oslo 1970 (Proc. Fifth Nordic Summer School in Math.), 291--313. Wolters-Noordhoff, Groningen, 1972.

\bibitem{H23} Hironaka, H., Gardening of infinitely near singularities. Algebraic geometry, Oslo 1970 (Proc. Fifth Nordic Summer School in Math.), 315-332. Wolters-Noordhoff, Groningen, 1972.

\bibitem{H24} Hironaka, H., The notion of near-complex subvariety, Notes by S. Nakano. Geometry of manifolds (Proc. Sympos., RIMS, Kyoto Univ., Kyoto, 1972), pp. 12-18. R.I.M.S. Sem. Rep. No. 158, Res. Inst. Math. Sci. Kyoto Univ., Kyoto, 1972.

\bibitem{H25} Hironaka, H., Introduction aux ensembles sous-analytiques, (French) R\'edig\'e par Andr\'e Hirschowitz et Patrick Le Barz. Singularit\'es \`a Carg\`ese (Rencontre Singularit\'es en G\'eom. Anal., Inst. Etudes Sci., Carg\`ese, 1972), pp. 13-20. Ast\'erisque, Nos. 7 et 8, Soc. Math. France, Paris, 1973.

\bibitem{H26} Hironaka, H., Subanalytic sets. Number theory, algebraic geometry and commutative algebra, in honor of Yasuo Akizuki, pp. 453-493. Kinokuniya, Tokyo, 1973.

\bibitem{H27} Hironaka, H., La vo\^ute \'etoil\'ee. (French) Singularit\'es \`a Carg\`ese (Rencontre Singularit\'es en G\'eom. Anal., Inst. Etudes Sci., Carg\`ese, 1972), pp. 415-440. Ast\'erisque, Nos. 7 et 8, Soc. Math. France, Paris, 1973.

\bibitem{H28} Hironaka, H., Lejeune-Jalabert, Monique, Teissier, Bernard, Platificateur local en g\'eom\'etrie analytique et aplatissement local, Singularit\'es \`a Carg\`ese (Rencontre Singularit\'es en G\'eom. Anal., Inst. Etudes Sci., Carg\`ese, 1972), pp. 441--463. Ast\'erisque, Nos. 7 et 8, 1973.

\bibitem{H29} Hironaka, H., Introduction to real-analytic sets and real-analytic maps, Quaderni dei Gruppi di Ricerca Matematica del Consiglio Nazionale delle Ricerche. Istituto Matematico "L. Tonelli" dell'Universit\`a di Pisa, Pisa, 1973. iii+162 pp.

\bibitem{H30} Hironaka, H., Introduction to the theory of infinitely near singular points. Memorias de Matem\'atica del Instituto "Jorge Juan", No. 28 Consejo Superior de Investigaciones Cient\'\i ficas, Madrid, 1974. vii+95 pp.

\bibitem{H31} Hironaka, H., Triangulations of algebraic sets. Algebraic geometry (Proc. Sympos. Pure Math. {\bf 29}, Humboldt State Univ., Arcata, Calif., 1974), pp. 165-185. Amer. Math. Soc., Providence, R.I., 1975.

\bibitem{H32} Hironaka, H., Flattening of analytic maps, Manifolds, Tokyo 1973 (Proc. Internat. Conf., Tokyo, 1973), pp. 313-321. Univ. Tokyo Press, Tokyo, 1975.

\bibitem{H33} Hironaka, H., Flattening theorem in complex-analytic geometry, Amer. J. Math. {\bf 97} (1975), 503-547.

\bibitem{H34} Aroca, Jos\'e M., Hironaka, H., Vicente, Jos\'e L. The theory of the maximal contact. Memorias de Matem\'atica del Instituto "Jorge Juan", No. 29. [Mathematical Memoirs of the ``Jorge Juan" Institute, No. 29] Instituto "Jorge Juan" de Matem\'aticas, Consejo Superior de Investigaciones Cient\'\i ficas, Madrid, 1975. 135 pp.

\bibitem{H35} Hironaka, H., Idealistic exponents of singularity, Algebraic geometry (J. J. Sylvester Sympos., Johns Hopkins Univ., Baltimore, Md., 1976), pp. 52-125. Johns Hopkins Univ. Press, Baltimore, Md., 1977.

\bibitem{H36} Hironaka, H., Stratification and flatness, Real and complex singularities (Proc. Ninth Nordic Summer School/NAVF Sympos. Math., Oslo, 1976), pp. 199-265. Sijthoff and Noordhoff, Alphen aan den Rijn, 1977.

\bibitem{H37} Hironaka, H., Bimeromorphic smoothing of a complex-analytic space. Acta Math. Vietnam. 2 (1977), no. 2, 103--168.

\bibitem{H38} Aroca, J. M., Hironaka, H., Vicente, J. L., Desingularization theorems, Memorias de Matem\'atica del Instituto "Jorge Juan" [Mathematical Memoirs of the Jorge Juan Institute], 30. Consejo Superior de Investigaciones Cient\'\i ficas, Madrid, 1977.

\bibitem{H39} Hironaka, H., On Zariski dimensionality type, Amer. J. Math. {\bf 101} (1979), no. 2, 384-419.

\bibitem{H40} Babakhanian, Ari; Hironaka, H., Formal functions over Grassmannians. Illinois J. Math. {\bf 26} (1982), no. 2, 201-211.

\bibitem{H41} Hironaka, H., On Nash blowing-up. Arithmetic and geometry, Vol. II, 103-111, Progr. Math., 36, Birkh\"auser, Boston, Mass., 1983.

\bibitem{H42} Hironaka, H. Urabe, Tohsuke, Kaiseki kukan nyumon. (Japanese) [Introduction to analytic spaces] Second edition. Suri Kagaku Raiburari [Mathematical Science Library], 1. Asakura Publishing Co. Ltd., Tokyo, 1983. vi+158 pp.

\bibitem{H43} Hironaka, H., Local analytic dimensions of a subanalytic set, Proc. Japan Acad. Ser. A Math. Sci. {\bf 62} (1986), no. 2, 73-75.

\bibitem{H44} Babakhanian, Ari; Hironaka, H., On the complex Grassmann
manifold. Illinois J. Math. {\bf 33} (1989), no. 1, 170-179.

\bibitem{H45} Hironaka, H., On the presentations of resolution data. Noted
and with an appendix by T. T. Moh. Algebraic analysis, geometry, and
number theory (Baltimore, MD, 1988), 135-151, Johns Hopkins Univ. Press,
Baltimore, MD, 1989.

\bibitem{H46} Hironaka, H., Fame, sweet and bitter. Miscellanea
mathematica, 155-176, Springer, Berlin, 1991.

\bibitem{H47} Hironaka, H., On the work of Shigefumi Mori. Proceedings
of the International Congress of Mathematicians, Vol. I, II (Kyoto, 1990),
19-25, Math. Soc. Japan, Tokyo, 1991.

\bibitem{H48} Connes, Alain, Flato, Mosh\'e, 
Hironaka, H.,
Jaffe, Arthur, Jones, Vaughan, Editorial (dedication of this issue to Huzihiro Araki). Comm. Math.
Phys. 155, (1993), no. 1,1--2.

\bibitem{H49} Catanese, Fabrizio., Hironaka, H., Kleiman, Steven L.,
et al. Obituary: Boris Moishezon. J. Algebraic Geom.
3 (1994), no. 4.
 
\bibitem{H50} Hironaka, H., Theory of infinitely near singular points.
J. Korean Math. Soc. {\bf 40} (2003), no. 5, 901-920.

\bibitem{H51} Hironaka, H., Three key theorems on infinitely near singularities.  Singularit\'es Franco-Japonaises, 87--126, S\'emin. Congr., 10, Soc. Math. France, Paris, 2005.
 
\end{thebibliography}

\title
{On the mathematical work of\\ Professor Heisuke Hironaka}

\author{ L\^e D\~ung Tr\'ang and Bernard Teissier} 
\maketitle
\textit{The authors wish to express their gratitude to Hei Hironaka for his wonderful teaching and his friendship, over a period of many years. }
\par\medskip
In this succinct and incomplete presentation of Hironaka's published work up to now, it seems convenient to use a covering according to a few main topics:  families of spaces and equisingularity, birational and bimeromorphic geometry, finite determinacy and algebraicity problems, flatness and flattening, real analytic and subanalytic geometry. This order follows roughly the order of publication of the first paper in each topic. One common thread is the frequent use of blowing-ups to simplify the algebraic problems or the geometry. For example, in the theory of subanalytic spaces of $\Rr^n$, Hironaka inaugurated and systematically used this technique, in contrast with the "traditional'' method of studying subsets of $\Rr^n$ by considering their generic linear projections to $\Rr^{n-1}$.\par
No attempt has been made to point at generalizations, simplifications, applications, or any sort of mathematical descent of Hironaka's work, since the result of such an attempt must be either totally inadequate or of book length.\par\medskip\noindent
$\bullet$ \textbf{Families of algebraic varieties and analytic spaces, equisingularity.}\par\medskip\noindent - Hironaka's first published paper is \cite{H1}, which contains part of his Master's Thesis. The paper deals with the difference between the arithmetic genus and the genus of a projective curve over an arbitrary field. In particular it studies what is today known as the $\delta$ invariant of the singularities of curves. Previous work in this direction had been done by Rosenlicht (in his famous 1952 paper where Rosenlicht differentials are introduced), as he points out in his review of Hironaka's paper in Math. Reviews. However, Rosenlicht's treatment is rather ''arithmetical'', in the style of Chevalley's book on algebraic functions of one variable, while Hironaka's presentation is ''geometrical''. It allows him to study the behavior of the arithmetic genus under specialization of a curve over a  (quasi-excellent) discrete valuation ring and to prove the best possible result in this direction, using Zariski's principle of degeneration.\par\noindent
In a seminar at IHES in the spring of 1968 Hironaka explained his ideas on how to relate the topology of  complex analytic singularities and the algebraic properties of the local rings, and stated in particular that the constancy of the Milnor number in an analytic family of plane curve singularities should imply its equisingularity. In this case the Milnor number is $2\delta -r+1$ where $r$ is the number of analytic branches. This was later proved by L\^e and Ramanujam for analytic families of hypersurfaces of dimension $\neq 2$ with isolated singularities. 
\par\medskip\noindent
- Next comes \cite{H2}, which gives necessary and sufficient conditions for the constancy of the Hilbert characteristic functions of the fibers of a family of projective varieties parametrized by the spectrum of a (quasi-excellent) discrete valuation ring. Again the approach is geometric. This paper contains the well known ''Lemma of Hironaka'' whose interest was pointed out by Nagata in his review in Math. Reviews. It gives a condition which ensures that the normality of the special fiber of such a family implies the normality of the total space.\par\medskip\noindent
- In the late 1960's, Zariski was developing the theory of equisingularity, and one problem was to understand the relationship between Zariski equisingularity, which certainly implies equimultiplicity, and the Whitney conditions. Hironaka used a geometric interpretation of his own ideas about normal flatness to give an analogue, \textit{normal pseudo-flatness}, which makes sense in both the real and the complex case, and to prove that the Whitney conditions along a stratum imply normal pseudo-flatness along that stratum. In the complex analytic case, normal pseudo-flatness implies equimultiplicity, so that the Whitney conditions imply equimultiplicity.
\par\medskip\noindent 
- In the paper \cite{H33} Hironaka proves the existence of Whitney stratifications for subanalytic sets (see below) and describes an algebraic condition (in terms of blowing-ups) ensuring Thom's $a_f$ condition for pairs of strata in $X$ with respect to a flat map $f\colon X\to S$. Thom's condition concerns the limiting positions of tangent spaces to the fibers of $f$ on the strata. It suffices to ensure a fairly good behavior of the fibers of $f$, for example the existence of vanishing cycles. Hironaka proves the existence of a stratification of $X$ satisfying Thom's condition whenever $S$ is a non singular curve.
\par\medskip\noindent 
- The paper \cite{H39} gives a proof of a fundamental result in equisingularity theory: the semicontinuity of Zariski's dimensionality type. Zariski had given an inductive definition of equisingularity of a hypersurface $X$ along a non singular subspace $Y$ which is based on equisingularity along the image of $Y$ of the discriminant of a \textit{generic} projection of $X$ to a non singular space of the same dimension. Here generic does not mean generic among linear projections; one must deal with formal projections given by $\hbox{\rm dim}X$ formal power series. The dimensionality type of a hypersurface at a point is also defined inductively; roughly speaking, $X$ is of dimensionality type $r$ at a point if it is equisingular at that point along a non singular subspace of codimension $r$ in $X$. The proof uses subtle constructions and a version of the Weierstrass preparation theorem.\par\medskip\noindent
$\bullet$ \textbf{Birational and bimeromorphic Geometry.}\par\medskip\noindent
- The first text is Hironaka's 1960 Harvard Thesis \textit{On the theory of birational blowing-up}. It presents a very complete view of the models of a field $K/k$ of algebraic functions, whether algebraic or ''Zariskian'' (corresponding to parts of the Zariski-Riemann variety of $K/k$). The main idea is to classify the models $V'$ which are projective over a given model $V$ according to the semigroup of those (coherent sheaves of fractional) ideals on $V$ which become locally principal on $V'$. More precisely, given a projective birational map $V'\to V$ Hironaka introduces an algebraic equivalence relation on the ideals on $V$ which is compatible with the product, shows that equivalent ideals are principalized by the same morphisms and defines $C(V,V')$ to be the commutative semigroup (for the operation induced by the product) of equivalence classes of ideals whose pull-back ideal is locally principal on $V'$. Then he shows that the smallest group $G(V,V')$ containing $C(V,V')$ is finitely generated and free. Thus $C(V,V')$ generates a convex cone in $G(V,V')\otimes_\Z\Qq$. This gives naturally rise to a combinatorial complex. The main theorem is that there is a one to one correspondence between the cells of this combinatorial complex and the normal models of $K/k$ which lie between $V$ and $V'$ and are projective over $V$. The inclusion of cells corresponds to birational domination of models. This is the first occurrence of the characteristic cone of a morphism. The thesis contains many results on the normalized blowing-up of ideals, the behaviour of the Picard group and the additivity of the depths of the base and fiber at a general point of the source of a nice morphism of schemes. (see also \cite{H3}). It also contains an example of a non projective birational map.
\par\medskip\noindent
- Hironaka produced in \cite{H4} (using a very ingenious blowing-up) a one-parameter family of non singular compact complex varieties whose special fiber is non-K\"ahler and all other fibers are K\"ahler.
\par\medskip\noindent
- The famous paper on resolution of singularities in characteristic zero (see \cite{H7}) introduced a number of new techniques in the subject, indeed in algebraic geometry and commutative algebra, as well as some key ideas which still underlie all the proofs of resolution of singularities in characteristic zero, after more than forty years and a lot of successful work to streamline and simplify Hironaka's proof and to make it effective.\par One of the simpler ideas he introduces is to measure the singularity of an algebraic variety $X$ at one of its points $x$ by the Hilbert-Samuel function instead of its multiplicity. It is the function $\hbox{\rm H}_{X,x}$ from $\N$ to $\N$ defined by: $n\mapsto \hbox{\rm dim}_{\kappa(x)}{\cal O}_{X,x}/m_{X,x}^{n+1}$ which is associated to the local ring ${\cal O}_{X,x}$. Asymptotically it behaves like $\hbox{\rm e}_{X,x}\frac{n^d}{d!}$ where $d$ is the dimension and $\hbox{\rm e}_{X,x}$ the multiplicity. Multiplicity is blind to lower-dimensional components so that multiplicity one does not imply non singularity in general. If the Hilbert-Samuel function of a local ring is that of a regular ring, the local ring is regular. Moreover, the Hilbert-Samuel function of the local ring of a variety determines its local embedding dimension, while the multiplicity does not. The partition of a variety according to the values taken by the Hilbert-Samuel function of the local rings (Samuel stratification of $X$) is of an algebraic nature. Next comes the fact that the constancy of the Hilbert-Samuel function of a variety $X$ along a non singular closed subvariety $Y$ (i.e., $H_{X,y}(n)$ is independant of $y\in Y$ for all $n\in \N$)  is equivalent to the flatness of the canonical map $C_{X,Y}\to Y$ of the normal cone of $Y$ in $X$ (normal flatness). This makes it possible to prove that under a \textit{permissible} blowing-up $f\colon X'\to X$, the Hilbert-Samuel function cannot increase, in the sense that for all $x'\in X'$ we have $\hbox{\rm H}_{X',x'}(n)\leq \hbox{\rm H}_{X,f(x')}(n)\ \ \forall n\in \N$. Here permissible blowing-up means a blowing-up of the ambient non singular space with a non singular center along which our singular space $X$ is normally flat. Moreover we have information on those points $x'\in f^{-1}(x)$ for which the two functions $\hbox{\rm H}_{X,x}$ and $\hbox{\rm H}_{X',x'}$ are equal.\par Now the key problem is, given a Hilbert-Samuel function $H$ of a point $x$ of $X$, to prove the existence of a sequence of permissible blowing-ups after which the Hilbert-Samuel function has decreased everywhere above a neighborhood of $x$. This suffices since one can show that the Hilbert-Samuel function cannot indefinitely decrease, and when it stabilizes with respect to a sequence of permissible blowing-ups, the space is non singular.\par
The key ideas introduced by Hironaka at this point are those of \textit{maximal contact} and of \textit{idealistic exponent}. They do not appear with these names in the Annals of Mathematics paper, but as parts of \textit{$J$-stable regular $\tau$-frames} and of \textit{resolution data of type} ${\cal R}_{II}^{N,n}$. Their role is explained in \cite{H35},\cite{H31},\cite{H37}. The idea is to find locally at each singular point $x$ of our singular space $X$ a non singular subspace $W$ of the ambient non singular space, which has ''maximal contact'' with $X$ at $x$ in the sense that it locally contains the Hilbert-Samuel stratum of $x$ in $X$, and after a permissible blowing-up its strict transform $W'$ contains all the points $x'\in f^{-1}(x)$ of the strict transform $X'$ of $X$ where $\hbox{\rm H}_{X',x'}= \hbox{\rm H}_{X,x}$ and has the same property at each of these points as $W$ at $x$. Given such a $W$, which exists in characteristic zero and has dimension $\leq \hbox{\rm dim}_xX$,  Hironaka then constructs an \textit{idealistic contact exponent} which in some sense measures the contact of $X$ with $W$ and how difficult it is to separate the strict transforms of $X$ and $W$ through permissible blowing-ups.\par An idealistic exponent is an equivalence class of pairs $(J,b)$ of an ideal on $W$ and an integer $b$. If we call $\nu_x(I)$ the order of an ideal $I$ with respect to the $m_{X,x}$-adic filtration, that is the largest $k$ such that $I_x\subset m^k_{X,x}$, the singular set of an idealistic exponent represented by a pair $(J,b)$ is the set $\{x\in X/ \nu_x(J)\geq b\}$. It is independant of the choice of the representative and an idealistic exponent is non singular if its singular locus is empty. There is a notion of permissible blowing-up for idealistic exponents and a rule for their transformation by such blowing-ups. Given $X$ and a $W$ with maximal contact as above, an idealistic contact exponent on $W$ has the following properties:\par\noindent $\bullet$ Blowing ups $f\colon X'\to X$ that are permissible for $X$ have their centers in $W$ and are permissible for $(J,b)$.\par\noindent $\bullet$ At all points $x'\in f^{-1}(x)$ of the strict transform $X'$ of $X$ where $\hbox{\rm H}_{X',x'}= \hbox{\rm H}_{X,x}$, the transform $(J', b')$ on the strict transform $W'$ of $W$ is again an idealistic contact exponent.\par\noindent $\bullet$ To make the Hilbert-Samuel function of $X$ drop by a sequence of permissible blowing-ups is equivalent to ''resolving the singularities'' of the idealistic exponent $(J,b)$ by a sequence of permissible (for $(J,b)$, but it is the same as for $X$) transformations on $W$.\par 
In an essential way, the dimension of the problem has dropped to the dimension of $W$, and Hironaka shows that resolution of singularities of spaces embedded in lower dimension than $X$ implies resolution for idealistic exponents on $W$, hence the existence of a sequence of permissible blowing-ups which makes the Hilbert-Samuel function of $X$ drop, hence resolution for $X$.\par\medskip
In the special case where $X$ is defined by one equation of the form $$Z^n+a_1(x_1,\ldots , x_k)Z^{n-1}+\cdots +a_n(x_1,\ldots , x_k)=0,$$ the space $W$ is the non singular space $Z+\frac{a_1(x_1,\ldots ,x_k)}{n}=0$ corresponding to the Tschirnhaus transformation: the change of the variable $Z$ to $$Z'=Z+\frac{a_1(x_1,\ldots ,x_k)}{n}$$ makes the term $a_1(x_1,\ldots , x_k)Z^{n-1}$ disappear. Assuming now that $a_1=0$, the idealistic contact exponent is $\bigl((a_i^{b/i})_{ 2\leq i\leq n}, b=n!\bigr)$. Note that if $k=1$ and $X$ is a plane curve, the rational number $\hbox{\rm min}_{1\leq i\leq n}\nu_x(a_i)/i$ is the slope of the first side of the Newton polygon of the curve. If the coordinates are chosen so that $Z=0$ has maximal contact, the integral part of this number measures how many blowing-ups (of points) one must make in order to make the multiplicity (or, which is equivalent in this case, the Hilbert-Samuel function) of the strict transform of the curve drop. This is an important part of the philosophy of the general proof, and also of the attempts to prove the positive characteristic case (see \cite{H11}, \cite{H12}, \cite{H13}, \cite{H50}, \cite{H51}).\par Note also that one sees immediately from the definition of the Tshirnhaus transformation why maximal contact might fail in positive characteristic, as indeed it does.\par In the general case, the construction of local systems of coordinates and equations which allow one to describe locally the Samuel stratum of the point and of the points obtained after permissible blowing-ups and to follow the behavior of idealistic exponents under permissible blowing ups at all the points where the Hilbert-Samuel function does not decrease requires a lot of work. In particular, if $X$ is defined locally at $x$ in a non singular space $T$, in order for local equations to contain enough information to describe locally the Hilbert-Samuel stratum, their initial forms (or leading forms) for the $m_{T,x}$-adic filtration of ${\cal O}_{T,x}$ must generate the ideal of the tangent cone $C_{X,x}$ (and similarly for the normal cone $C_{X,Y}$ along a permissible center): this is the foundation of the notion of \textit{standard basis}, closely related to what is called nowadays Gr\"obner basis or Macaulay basis.\par\medskip
The problem of resolution of singularities for complex analytic spaces, about which Hironaka explained his first ideas in a seminar in Ruuponsaari (Finland) in the summer of 1968, and which was studied in \cite{H21}, \cite{H23}, \cite{H30}, \cite{H34}, \cite{H37} (a survey of the method), and \cite{H38}, presents entirely new difficulties.\par\medskip\noindent
- Starting from an ideal generated by holomorphic functions, in order to find useful generators for the ideal $I$ of the idealistic exponent, one must first apply a generalized form of the Weierstrass preparation theorem, invented by Hironaka for this purpose, and which will appear again in the paragraph devoted to flattening. It is in fact rather a division theorem, dividing an element $g$ in a convergent power series ring $\C\{z_1,\ldots, z_N\}$ by the generators $f_1,\ldots, f_\ell$ of an ideal $I$, that is, writing $g=h_1f_1+\cdots +h_\ell f_\ell+r$ in such a way that no monomial $z^A$ of $r$ is divisible by the initial monomials of the series $f_i$. It is closely related to a theorem of Grauert (1972).\par\medskip\noindent
- The centers of blowing-up which one can construct locally for the analytic topology do not necessarily extend as global centers of blowing-up as the centers constructed locally for the Zariski topology do in algebraic geometry. To overcome this difficulty, Hironaka invented the \textit{Gardening of infinitely near singularities} explained in \cite{H23} and in the Madrid notes \cite{H30}. It is a sort of sheafification of local sequences of permissible blowing-ups which can be used to prove by induction on the dimension the existence of global centers of permissible blowing-ups. \par\medskip\noindent
More recently (see \cite{H50}, \cite{H51}) Hironaka has taken up again the proof of resolution of singularities of excellent schemes in arbitrary characteristic. He has introduced a graded algebra which is finitely generated in any characteristic and is associated to sequences of permissible blowing-ups. In principle the finite-generatedness of this algebra and its very special properties with respect to differential operators and some restrictions of ambient space shall make up for the absence of maximal contact in positive characteristic. This is work in progress.\par\medskip\noindent
- Hironaka has also made a substantial contribution to the problem of resolution of singularities by Nash modification. The Nash modification $\nu\colon N(X)\to X$ of an algebraic variety or a reduced equidimensional complex analytic space is the minimal proper birational (or bimeromorphic) map such that $\nu^*\Omega^1_X$ has a locally free quotient of rank $\hbox{\rm dim}X$. The fibers $\nu^{-1}(x)$ are set-theoretically the limit directions at $x$ of tangent spaces to $X$ at non singular points tending to $x$. The problem of proving that after finitely many Nash modifications one obtains a non singular space was first stated by Semple in 1954, and essentially no progress was made until Hironaka proved in \cite{H41} by a valuative argument that if one considers the same problem for surfaces and with \textit{normalized} Nash modification instead of Nash modification, one could reduce to the case where $X$ has only rational singularities.
\par\medskip
\noindent
- Another important contribution of Hironaka to bimeromorphic geometry is the definition in \cite{H27} and \cite{H29} of the complex-analytic analogue of the Zariski-Riemann manifold, the\textit{Vo\^ute \'etoil\'ee}. Instead of taking the projective limit of birational blowing-ups, one considers compositions of local blowing-ups. A local blowing-up $X'\to X$ of $X$ is the blowing-up in an open subset $U$ of $X$ of a closed analytic subset of $U$, composed with the inclusion $U\subset X$. It is not proper, and compositions of local blowing-ups do not form a projective system. An element of the Vo\^ute \'etoil\'ee ${\cal E}_X$ is a maximal projective subsystem of the system of all compositions of local blowing-ups above $X$, satisfying in addition some technical condition. There is a canonical proper morphism ${\cal E}_X\to X$. This object renders the same services as the Zariski-Riemann manifold, and in particular is crucial in the proof of the local flattening theorem below.
\par\medskip
\noindent 
- In \cite{H14} it is shown that on a non singular projective variety $X$ of dimension $n$ every algebraic cycle of dimension $d\leq \hbox{\rm min}(3,\frac{n-1}{2})$ is rationally equivalent to a linear combination of non singular subvarieties of $X$. The method is to desingularize, move into general position in the blown-up space, and push down. The first paragraph contains the theorem that a proper flat morphism of noetherian schemes with smooth fibers which has a fiber which is a projective space is a projective bundle.
\par\medskip
\noindent
- Under the pseudonym of Hej Iss'sa (Kobayashi Issa is a famous Japanese poet) Hironaka settled in 1966 (see \cite{H10}) a long standing problem in the theory of Riemann surfaces. Here we consider, for non compact connected Riemann surfaces, the ring $A(X)$ of holomorphic functions on $X$, and its field of fractions $F(X)$. A famous theorem of Chevalley-Kakutani and Bers showed that $X$ and $Y$ are conformally equivalent if and only if $A(X)$ and $A(Y)$ are isomorphic as $\C$-algebras. Iss'sa proved that $X$ and $Y$ are conformally equivalent if and only if the fields $F(X)$ and $F(Y)$ are $\C$-isomorphic. The argument is in fact valid in a much more general setting. The key fact is that a valuation of the field $M(X)$ of meromorphic functions on a complex variety $X$ whose value group is "not too large" in a precise sense, and in particular not divisible, is necessarily $\geq 0$ on the ring of holomorphic functions on $X$. The crucial case is the case $X=\C$. \par\medskip\noindent
\par\medskip\noindent
$\bullet$  \textbf{Finite determinacy, algebraicity.}\par\medskip\noindent
The general result stated (but proved only in special cases) in \cite{H9} is that given a finite type flat map $\pi\colon X\to Y$ of schemes, which, over a neighborhood of a point $y_0\in Y$, has reduced equidimensional fibers, all of the same dimension and is endowed with a section $\epsilon \colon Y\to X$, and given a closed subscheme $Z$ of $X$ containing $\epsilon (Y)$ and such that $\pi\vert X\setminus Z\colon X\setminus Z\to Y$ is formally smooth, then there exists an $H$-adic $(t,r)$-index for $(Y,y_0,\pi, X,\epsilon)$, where $H$ is the ideal defining $Z$ in $X$. The completion of $X$ along $Z$ is defined as the completion of the sheaf of algebras of $X$ with respect to the $H$-adic topology. It is the inductive limit of the infinitesimal neighborhoods of $Z$ in $X$, which are defined by the powers of $H$. Roughly speaking, the existence of a $(t,r)$-index means that, if the dimensions of the fibers are fixed, the completion of $X$ along $Z$  is, in a neighborhood of $\epsilon (y_0)$ and up to $Y$-isomorphism, determined by a finite infinitesimal neighborhood of $Z$ in $X$.  There is a subtlety in that, if we assume that we have two data $(Y,y_0,\pi, X,\epsilon, Z)$ and $(Y,y_0,\pi', X',\epsilon',Z')$ with the same fiber dimension, it is not an isomorphism of the $t$-th infinitesimal neighborhood $Z_t$ of $Z$ in $X$ with an infinitesimal neighborhood $Z'_t$ of $Z'$ in $X'$, which extends to the completions, but its restriction to some $Z_{t-r}$.\par
Hironaka writes at the end of \cite{H9} that one of his sources of inspiration was a result of Grauert stating that the formal completion along the exceptional divisor of a point-blowing-up is determined by a finite infinitesimal neighborhood. Generalizing this to exceptional divisors with normal crossings (see \cite{H6}) and applying resolution of singularities, one can hope to prove for example that any isolated singularity is analytically or formally determined by a finite infinitesimal neighborhood of the singular point, and in particular is algebraic, but there are difficulties. In any case, for complete intersections with isolated singularities, this result is proved in \cite{H9}, and the determinacy of a small complex neighborhood by a finite infinitesimal neighborhood for an exceptional divisor $A$ in a complex space $X$ such that $X\setminus A$ is non singular is proved in \cite{H6}.\par
The papers \cite{H15}, \cite{H16}, \cite{H17} all deal with the problem of comparing formal or analytic structures (embeddings, line bundles, etc.) with algebraic or rational ones. 
 
\par\medskip\noindent
$\bullet$  \textbf{Flatness and Flattening.}\par\medskip\noindent
- In his study of algebraicity conditions and the $(t,r)$-index (\cite{H9}) Hironaka used in the 1960's blowing-ups to transform an arbitrary subscheme $X$ of a non singular ambient scheme, defined by an ideal $I$, into a local complete intersection, which amounts essentially to making the (strict transform of the) normal space corresponding to the coherent sheaf of ${\cal O}_X$-modules $I/I^2$ flat. This can be systematized in at least two ways in analytic geometry. The main idea is that given any analytic map $f\colon X\to S$ it can be made flat by base change  in the following sense: \par\noindent
- If the map $f$ is proper and $S$ is reduced and countable at infinity, given any coherent sheaf on $X$, there exists a proper bimeromorphic map $\pi\colon S'\to S$ such that in the diagram
$$\begin{array}{lllrrr} 
X\times_SS'&\stackrel{\pi'}\longrightarrow & &X\\
f'\downarrow&&&f\downarrow 
\\
\ \ \ S'&\stackrel{\pi}\longrightarrow  &&S\\
\end{array}$$
the sheaf $\pi'^*F$ modulo its $f'$-torsion is $f'$-flat.\par\noindent
- We no longer assume $f$ to be proper and consider a point $s$ of $S$ and a compact subset $L\subset f^{-1}(s)$. Then there exists a finite family of maps $\pi_k\colon S'_k\to S$, each of which is a finite composition of local blowing-ups, such that the strict transform of $f$ by each $\pi_k$ is flat at every point mapped to a point of $L$. Here the strict transform is the closure of the part of the fiber product whose image by the projection to $S'_k$ is not contained in the exceptional divisor of $\pi_k$. Moreover, the union of the images of the maps $\pi_k$ is a neighborhood of $s$ in $S$. 
\par
In both cases, the first difficulty is to define a \textit{flattener} of the sheaf (or the map). The proof of flattening in the proper case is given in \cite{H33} (see also \cite{H32}). In the local case, the flattener is the largest closed subspace $B$ of $S$ \textit{containing the point} $s$ such that the restriction of $f$ above $B$ (in the sense of fiber product with $B$ over $S$) is flat.\par
The first proof of local flattening in \cite{H28} used generalized Newton polygon techniques. There is in \cite{H29} a proof of the existence of the flattener using a parametrized version (with respect to $S$) of Hironaka's division theorem for the local equations of $X$ in $S\times \C^N$.\par In particular, if $A\to B=A\{t_1,\ldots, t_n\}/I$ is a flat morphism of complex-analytic algebras, Hironaka's division theorem provides, given a system of generators of $I$ and after a suitable linear change of the coordinates $t_1,\ldots, t_n$, an explicit presentation of the $A$-module $B$ as a finite direct sum of free $A$-modules of the form $A\{t_1,\ldots, t_j\}$. If $B$ is not flat over $A$, the division theorem makes the obstructions to this presentation appear as series in $A\{t_1,\ldots, t_n\}$, whose coefficients in $A$ provide equations for the flattener of the corresponding map.\par
It should be pointed out that flattening has important consequences in birational and bimeromorphic geometry. For example if $f\colon X\to Y$ is a proper bimeromorphic map of complex spaces, after base change by a blowing-up $Y'\to Y$ the induced map (strict transform) $f'\colon X'\to Y'$ is bimeromorphic and flat, so it is an isomorphism. Inverting it we see that $f$ is dominated by a blowing-up, a form of Chow's lemma.  \par\medskip\noindent
$\bullet$  \textbf{Real analytic and subanalytic Geometry; subanalytic sets.}\par\medskip\noindent
- In the 1970's Hironaka developed his theory of subanalytic sets. The existence and usefulness of such a theory had been foreseen by Thom and \L ojasiewicz in the 1960's, and in 1968 Gabrielov proved the theorem of the complement: if we call a subset of $\Rr^n$ subanalytic when it is locally at every point of $\Rr^n$ a finite union of differences of images of semi-analytic sets by proper analytic maps, the complement of a subanalytic set is subanalytic. From there the theory could be developed, and indeed was later, from the viewpoint of projections of subsets of $\Rr^n$ to affine spaces of lower dimensions introduced by \L ojasiewicz in the study of semi-analytic sets.\par
In \cite{H25}, \cite{H26} and \cite{H29}(which contains a complete exposition), Hironaka built the whole theory from a different viewpoint, as explained in the beginning. The main theorem is a resolution of singularities of subanalytic sets: Every subanalytic set is locally the finite union of images of spheres by real-analytic maps. This is a consequence of the \textit{Rectilinearization theorem} which states that a subanalytic set can locally be transformed by finitely many finite sequences of local blowing-ups into a union of quadrants in affine space.  One important step in the proof is the use of the local flattening theorem in complex analytic geometry to the complexifications of suitable real analytic maps defining the subanalytic set to prove that after suitable blowing-ups of the ambient space a subanalytic set becomes semi-analytic. Then one can apply resolution of singularities to the definining analytic functions.\par
In \cite{H31} Hironaka gives a proof of the triangulability of semi-algebraic sets (a known theorem, reputedly difficult) which is so streamlined and clear that it allows him, almost without change, to prove the triangulability of subanalytic sets, a new theorem. Finally by 1976 Hironaka had provided a complete analytic description of subanalytic sets and their finiteness properties, including Whitney stratifications, triangulability and the \L ojasiewicz inequalities.\par\medskip
\textit{We are grateful to Herwig Hauser and Tadao Oda for their careful reading and their suggestions}

\end{document} 

\end{document}